 Vc/16-Nov-04
Week47 starting 15Nov 04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 15 Nov 04  ~12.20 hrs 

	LHC SM18 Magnet Test Facility 


	10
	A1
	MBBL3220
	ICS 4 Final connection
	295.2

	3
	A2
	MBBL2112
	PT 2 HV Cold Test
	1.9

	5
	B1
	SSS042
	PT 10.1: Prov. Quench 1/2 HF 11850
	11.6

	6
	B2
	MBBR3241
	PT 12 Warm Up
	267.4

	2
	C1
	MBAR1138
	PT 6.1 Training 1
	1.9

	8
	C2
	MBBL2119
	Prep 5 Pumping + Cool Down
	205.4

	9
	D1
	SSS039
	ICS 4 Final connection
	294.5

	1
	D2
	SSS071
	Survey @ Cold SC14
	1.91

	12
	E1
	MBBL1164
	Finished Ready to go Final scan
	298

	4
	E2
	MBAL2097
	PTC 9 HV @ Cold LAST
	1.9

	7
	F1
	MBBL1147
	Prep 5 Pumping + Cool Down
	2.35

	11
	F2
	MBBR3244
	ICS 4 Final connection
	293.8


11/15/2004 12:21:40 PM    Temperature data are taken from TBSM18 site
Coordinator this week: Vittorio; MTM support on call :   16-3602 .

12 named  Magnets Tested or 13 Magnets tests :

 10 dipoles( 2 with MM & long MM)+ 2 SSS(1 with MM) +  0  SSSS     Disconnected after tests last week ( Sat Midnight to Sat midnight) 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
SSW system & Guy 

AOB

Tests 

	Magnets under Test from 07-11-2004 to 13-11-2004 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBBL2098 
	[image: image1.png]



	31-10-2004 
	28 h.
	167 h.
	32 h.
	15 h.
	7 h.
	10-11-2004 
	240 hours
	
	

	2
	TBA2
	MBBL3219 
	[image: image2.png]



	29-10-2004 
	24 h.
	44 h.
	160 h.
	18 h.
	0 h.
	09-11-2004 
	264 hours
	
	

	3
	TBA2
	MBBL3219 
	[image: image3.png]



	09-11-2004 
	7 h.
	51 h.
	12 h.
	19 h.
	4 h.
	11-11-2004 
	48 hours
	
	

	4
	TBB2
	SSS069 
	[image: image4.png]



	01-11-2004 
	57 h.
	31 h.
	51 h.
	13 h.
	13 h.
	08-11-2004 
	168 hours
	
	

	5
	TBC1
	MBBR3243 
	[image: image5.png]



	06-11-2004 
	29 h.
	28 h.
	15 h.
	23 h.
	9 h.
	11-11-2004 
	120 hours
	
	

	6
	TBC2
	MBBL2113 
	[image: image6.png]



	04-11-2004 
	41 h.
	20 h.
	15 h.
	13 h.
	6 h.
	08-11-2004 
	96 hours
	
	

	7
	TBD1
	SSS075 
	[image: image7.png]



	05-11-2004 
	18 h.
	32 h.
	76 h.
	19 h.
	13 h.
	12-11-2004 
	168 hours
	
	

	8
	TBE1
	MBBL2116 
	[image: image8.png]



	05-11-2004 
	15 h.
	36 h.
	18 h.
	20 h.
	7 h.
	09-11-2004 
	96 hours
	
	

	9
	TBE2
	MBAL1161 
	[image: image9.png]



	03-11-2004 
	28 h.
	23 h.
	15 h.
	16 h.
	10 h.
	07-11-2004 
	96 hours
	
	

	10
	TBE2
	MBAL2097 
	[image: image10.png]



	07-11-2004 
	27 h.
	27 h.
	51 h.
	20 h.
	0 h.
	12-11-2004 
	120 hours
	
	

	11
	TBF1
	MBAL2092 
	[image: image11.png]



	07-11-2004 
	40 h.
	7 h.
	16 h.
	19 h.
	14 h.
	11-11-2004 
	96 hours
	
	

	12
	TBF1
	MBBL3218 
	[image: image12.png]



	01-11-2004 
	27 h.
	27 h.
	57 h.
	15 h.
	8 h.
	07-11-2004 
	144 hours
	
	

	13
	TBF2
	MBBL3221 
	[image: image13.png]



	02-11-2004 
	24 h.
	69 h.
	29 h.
	15 h.
	5 h.
	08-11-2004 
	144 hours
	
	


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL2098
	11778 A.
(D2_U)
	12606 A.
(A2-CF-S3)
	10753 A.
(D2-U)
	11081 A.
(D2-U)
	10 A.
(No Data)
	10966 A.
(D2_U)
	11054 A.
(A2 CF S3)
	10980 A.
(D2_U)
	[image: image14.png]



	-
	-
	-
	-

	2
	MBBL3219
	10660 A.
(D2-L-S01)
	10417 A.
(D1-U-S01)
	11741 A.
(D2_L_S01)
	12460 A.
(D2_U_S01)
	12779 A.
(D@_U_S01)
	12365 A.
(D2-U-?)
	12345 A.
(D2-U)
	12654 A.
(D2 U)
	12740 A.
(D2-U)
	[image: image15.png]



	-
	-
	-

	3
	MBBL3219
	10791 A.
(D1-L)
	11631 A.
(D2_L_MR)
	12696 A.
(D2_L)
	[image: image16.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS069
	12250 A.
(QE_12)
	12248 A.
(QI_34)
	[image: image17.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBR3243
	12369 A.
(D2_U)
	12849 A.
[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBL2113
	11916 A.
(D2-L)
	12850 A.
[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS075
	12248 A.
(QI_12)
	12649.6 A.
(QE-34)
	[image: image20.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBBL2116
	10789 A.
(D1-L)
	12435 A.
(D1-U)
	[image: image21.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBAL1161
	12030 A.
(D1-U)
	12327 A.
(D2_L)
	[image: image22.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBAL2097
	11903 A.
(D2-U)
	12289 A.
(D1-U)
	12532 A.
(D1_L)
	12857 A.
[image: image23.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAL2092
	11410 A.
(D1-U-MRB)
	12708 A.
(D2-L MRB)
	12850 A.
[image: image24.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBL3218
	10289 A.
(No Data)
	10402 A.
(D2-L-S01)
	12266 A.
(D2 L)
	12491 A.
(D2-L)
	12850 A.
[image: image25.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBBL3221
	9207 A.
(D1-L)
	10855 A.
(D2-L)
	11525 A.
(D1-U)
	12773 A.
(D1-L)
	12523 A.
(D1_L)
	12850 A.
[image: image26.jpg]



	-
	-
	-
	-
	-
	-
	-


	From 07-11-2004 to 13-11-2004 

	-
	Bench
	Magnet Name
	shafts
	Start Date .
	Connect
Time
	Wait
+Cooling
	Cold
Tests
	Wait 
+Warming
	Disconnect
Time
	End Date .
	Total
	Q

	1
	TBF2
	MBBL3221 
	False 
	02-11-2004 
	24 h.
	69 h.
	29 h.
	15 h.
	5 h.
	08-11-2004 
	6 days
	5 

	2
	TBF1
	MBAL2092 
	False 
	07-11-2004 
	40 h.
	7 h.
	16 h.
	19 h.
	14 h.
	11-11-2004 
	4 days
	2 

	3
	TBF1
	MBBL3218 
	True 
	01-11-2004 
	27 h.
	27 h.
	57 h.
	15 h.
	8 h.
	07-11-2004 
	6 days
	4 

	4
	TBE2
	MBAL1161 
	False 
	03-11-2004 
	28 h.
	23 h.
	15 h.
	16 h.
	10 h.
	07-11-2004 
	4 days
	2 

	5
	TBE2
	MBAL2097 
	False 
	07-11-2004 
	27 h.
	27 h.
	51 h.
	20 h.
	0 h.
	12-11-2004 
	5 days
	3 

	6
	TBE1
	MBBL2116 
	False 
	05-11-2004 
	15 h.
	36 h.
	18 h.
	20 h.
	7 h.
	09-11-2004 
	4 days
	2 

	7
	TBD1
	SSS075 
	True 
	05-11-2004 
	18 h.
	32 h.
	76 h.
	19 h.
	13 h.
	12-11-2004 
	7 days
	2 

	8
	TBC2
	MBBL2113 
	False 
	04-11-2004 
	41 h.
	20 h.
	15 h.
	13 h.
	6 h.
	08-11-2004 
	4 days
	1 

	9
	TBC1
	MBBR3243 
	False 
	06-11-2004 
	29 h.
	28 h.
	15 h.
	23 h.
	9 h.
	11-11-2004 
	5 days
	1 

	10
	TBB2
	SSS069 
	False 
	01-11-2004 
	57 h.
	31 h.
	51 h.
	13 h.
	13 h.
	08-11-2004 
	7 days
	2 

	11
	TBA2
	MBBL3219 
	True 
	29-10-2004 
	24 h.
	44 h.
	160 h.
	18 h.
	0 h.
	09-11-2004 
	11 days
	9 

	12
	TBA2
	MBBL3219 
	True 
	09-11-2004 
	7 h.
	51 h.
	12 h.
	19 h.
	4 h.
	11-11-2004 
	2 days
	3 

	13
	TBA1
	MBBL2098 
	False 
	31-10-2004 
	28 h.
	167 h.
	32 h.
	15 h.
	7 h.
	10-11-2004 
	10 days
	8 


 

from 07-11-2004 to 13-11-2004
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	Date
	Bench
	Problem
	Type
	Appl.
	Action/advise

	Tuesday _09_N


	TBE1
	Change of type for the magnet 1164 from type L to type R. Name not change in the bare code. Scan not correct
	MTM
	
	Magnet name entered manually.

	Tuesday _09_N


	TBD2
	Wire broken during rot-wire measurements


	MTM
	
	XY axis + test repeated.

	Wednesday _10_M


	TBA1
	Server crashed while using TEMA

Disk full.
	AB/CO
	
	TEMA re-launched.



	Wednesday _10_M


	TBD1
	LF DAQ not working.

SPA occurred.
	AB/CO
	LF
	

	Wednesday _10_N


	TBE2
	1.5 kA provoked quench (all HF). Alert jumpv>1.0.
	MTM
	AQA
	Maryline, TC contacted. They decided to repeat Resistance measurements and HV test. OK. Proceed  with the first training

	Wednesday _10_N


	TBE2
	TEMA blocked because of HWREC 
	AB/CO
	HWREC
	Maryline called. She gave the password to put HWREC off.

	Wednesday _10_N


	TBB1
	Powering EF circuit not possible. Inductance not corresponding to 16 mH
	AB/PO
	
	Test went one with training. Maryline changed the inductance values.

	Thursday _11_M
	
	Net work problem. AFS server is not working.
	IT
	
	Wait for re-start of AFS server.

	Thursday _11_M
	TBE2


	First quench at 12000 A. Oscillations in the direct signals. 
	MTM
	
	Stephane and Gijs contacted Pierre. He asked for another quench before investigation. Training pursued up to 9 T + thermal cycle.

	Thursday _11_N

Friday_12_N
	TBC2
	During charge of QH, QH1&2 trigger and discharge
	MTM
	
	Maryline informed. She will check the inter-out card.

	Thursday _11_N
	TBD
	The power supply 18 V is missing
	MTM
	
	Borrow that from TBB.

	Friday_12_M
	TBF
	HV test execution slow.
	AB/CO
	HV
	Alessandro found the raison: Bug due to a too big log file. Corrected also in MMP

	Friday_12_N
	TBB1
	MQT powering (EF). Ramp to 550 A but display of power showed wrong inductance value (MO).
	AB/CO
	
	Wrong name in the power supply  controller. Action Maryline to trigger the changes.

	Friday_12_N
	TBB1
	SSW measurement: Difficulty to fix the new wire: the washers on the tension motor spindle did not grip the wire tight enough. Slip of the wire when the tension increases
	MTM
	
	Action Guy Monday morning.

	Saturday_13_M
	TBA1
	Cryo loss on MB1134 during 1,5 kA (HF) provoked quench
	ACR
	
	

	Saturday_13_M
	TBF2
	Current reading in Power Application in TEMA not working during ramping.
	MTM
	Power
	

	Saturday_13_M
	TBB1
	Sun workstation (menu selection) not working correctly.
	AB/CO
	
	Adrian called. Disk full. He found some place.

	Saturday_13_M
	TBF2
	Bench selections rotary switch not working.
	AB/PO
	
	.Maryline contacted. IDCCT crashed. Problem solved by putting on the DCCT.

	Saturday_13_A
	TBD2
	PLC-SE blocked.

SSW measurement: Execution blocked when SSW window appeared. 

The same in the system installed on TBB1
	
	
	Maryline informed. Config file corrupted. Martin copied correct file from another measurements.



	Sunday_14_M
	TBD2
	Powering EF (MQT) circuit not possible (SSS071). 2 triggers are coming.
	MTM
	
	 Powered with 2 A/s instead of 5 A/s but with 16 m H of inductance.

Maryline contacted.

	Sunday_14_N
	TBE2
	Lf application did not create correct config.file

Problem with extension of training quench. Extension has to be changed from .a to .u
	AB/CO
	lf
	Maryline called. She advised to copy correct one from another test file.

	Sunday_14_N
	TBB1
	SSS042  did not reach 12850 A after 9 ramps.
	MTM
	
	TC decided to do 4.4 K quench before thermal cycle. Quench in the clamp at 6 kA. SS limit study at 4.4 K on SSS not possible because main leads not in He liquid.

	Monday_15_A
	TBC1
	MB1138: Anomalous decrease of I magnet.
	MTM
	
	Origin not known. Pierre informed.

	Monday_15_A
	TBC1
	Difficulties to recover the cryo OK at 1.9 K
	ACR
	
	Oil leak in the pumping group.


 









